Author's response to reviews

Title: Meta-analysis of adverse health effects due to air pollution in Chinese populations

Authors:

Hak Kan Lai (laihk@hku.hk)
Hilda Tsang (laihk@hku.hk)
Chit Ming Wong (laihk@hku.hk)

Version: 4 Date: 5 April 2013

Author's response to reviews:

Dear Editor,

We have revised our manuscript's abstract conclusion as advised. We have deleted the first sentence of the conclusion in the abstract. We look forward to hearing from you soon.

Yours sincerely,
Hak-Kan Lai